Author's response to reviews

Title: How to maintain the oral health in a Wolff Parkinson White Syndrome child: A case report

Authors:

Tsampikos Petroniatis (tsalaquila@hotmail.com)
Eleonora Ortu (eleortu@gmail.com)
Nicola Marchili (nic.marc@hotmail.it)
Mario Giannoni (mario.giannoni@cc.univaq.it)
Annalisa Monaco (annalisamonaco@yahoo.it)

Version: 3 Date: 24 June 2014

Author's response to reviews: see over
Title: How to maintain the oral health in a Wolff Parkinson White Syndrome child: A case report

Version: 2 Date: 16 June 2014

Reviewer: Giovambattista GD Desideri

Which of the following best describes what type of case report this is?: Other

If other, please specify:

The paper described the oral health management in a young female with Wolff Parkinson White

Has the case been reported coherently?: Yes

Is the case report authentic?: Yes

Is the case report ethical?: Yes

Is there any missing information that you think must be added before publication?: No

Is this case worth reporting?: Yes

Is the case report persuasive?: Yes

Does the case report have explanatory value?: Yes

Does the case report have diagnostic value?: Yes

Will the case report make a difference to clinical practice?: Yes

Is the anonymity of the patient protected?: Yes

Comments to authors:

The management of the oral health in this kind of patient is very important. I think this manuscript can help the specialist to adopt the correct procedures. I believe that the paper underlines the importance of the clinical diagnosis and treatments for this patients. I suggest to add the differential diagnosis with other cardiac
conditions. I suggest to add also the clinical manifestations of the pathology.

We added the differential diagnosis with other cardiac diseases in the discussion (highlighted in yellow). You can find also the clinical signs of the syndrome. We added other photos of ECG.

Level of interest: An article of importance in its field Quality of written English: Acceptable

Declaration of competing interests:

No competing interests to declare

Reviewer's report

Title: How to maintain the oral health in a Wolff Parkinson White Syndrome child: A case report

Version: 2 Date: 9 June 2014

Reviewer: Maurizio D'Amario

Which of the following best describes what type of case report this is?: Presentations, diagnoses and/or management of new and emerging diseases

Has the case been reported coherently?: Yes

Is the case report authentic?: Yes

Is the case report ethical?: Yes

Is there any missing information that you think must be added before publication?: Yes

Is this case worth reporting?: Yes

Is the case report persuasive?: Yes

Does the case report have explanatory value?: Yes

Does the case report have diagnostic value?: Yes

Will the case report make a difference to clinical practice?: Yes

Is the anonymity of the patient protected?: Yes
Comments to authors:

It's an interesting article. The manuscript well describes a Parkinson White Syndrome case in a child and its oral management, but some improvements are needed.

I would suggest to improve the article citing all the drugs used to manage this disease. Please also try to describe all the signs and symptoms of the Wolff Parkinson White Syndrome in the discussion section.

Check the correctness of the reference n. 10.

**We Corrected the reference n 10. You can find also the signs and symptoms of the disease in the introduction. There are the drugs used in the pathology in the INTRODUCTION.** The changes are highlighted in yellow.

Level of interest: An article of importance in its field

Quality of written English: Acceptable

Declaration of competing interests:

I declare that I have no competing interests